Author’s response to reviews

Title: Association of Retinol Binding Protein 4 and Transthyretin with Triglyceride Levels and Insulin Resistance in Rural Thais with High Type 2 Diabetes Risk

Authors:
Karunee Kwanbunjan (karunee.kwa@mahidol.ac.th;phosat_cr@hotmail.com)
Pornpimol Panprathip (lloook_kaew@hotmail.com)
Chanchira Phosat (chanchira.pho@mahidol.ac.th)
Noppanath Chumpathat (nophc@hotmail.com)
Naruemon Wechjakwen (Naruemon_w@hotmail.com)
Somchai Puduang (Somchai.pou@mahidol.ac.th)
Ratchada Auuyuenyong (ratchada_auy@hotmail.com)
Ina Henkel (ina.henkel@uni-potsdam.de)
Florian Schweigert (florian.schweigert@uni-potsdam.de)

Version: 3 Date: 30 Apr 2018

Author’s response to reviews:

April 30, 2018

Dear Prof. Francesca Dickens

Thank you for consider our research work. We carefully corrected some essential editorial revision.

1. In the Ethics approval and consent to participate section of the Declarations, we confirm that informed consent is written, obtained signatures from all participants. It was already clearly state in the manuscript (page 11, line 257).
2. The figure titles was already placed under heading 'Figure Legends' after the References (page 17, line 402).

3. The emails of authors were already placed on title page, (page 1, line 20).

4. The manuscript is clean version.

We looking forward to hearing from you soon. Thank again for your kindness.

Best Regards

Karunee Kwanbunjan